Author's response to reviews

Title: Cigarette smoke increases cardiomyocyte ceramide accumulation and inhibits mitochondrial respiration

Authors:

Trevor S. Tippetts (trevorstippetts@gmail.com)
Duane R. Winden (dwinden@roseman.edu)
Adam C. Swensen (mendriveminis@gmail.com)
Michael B. Nelson (michaelbrucenelson@gmail.com)
Mikayla O. Thatcher (mikaylathatcher@gmail.com)
Rex R. Saito (rexsaito30@hotmail.com)
Tyler B. Condie (tylercondie@gmail.com)
Kurtis J. Simmons (kurtis0911@gmail.com)
Allan M. Judd (Allan_Judd@byu.edu)
Paul R. Reynolds (paul_reynolds@byu.edu)
Benjamin T. Bikman (benjamin_bikman@byu.edu)

Version: 2
Date: 7 October 2014

Author's response to reviews:

1. We have replaced the word 'Introduction' with 'Background' at the beginning of the manuscript.
2. We have a statement in the 'Methods' section stating our approved use through the institutional IACUC to use animals.
3. We have added a 'Conclusions' section at the end of the discussion.
4. We have added a statement of competing interests (none).
5. We have added an 'Acknowledgements' section (none).
6. We have eliminated the redundant figure legends and placed them at the end of the manuscript.
7. We will upload figures individually.